Author's response to reviews

Title: Neck exercises, physical and cognitive behavioural-graded activity as a treatment for adult whiplash patients with chronic neck pain: Design of a randomised controlled trial

Authors:

Inge Ris Hansen (iris@health.sdu.dk)
Karen Søgaard (ksogaard@health.sdu.dk)
Robin Christensen (Robin.Christensen@frh.regionh.dk)
Bente Thomsen (bente.thomsen@slb.regionsyddanmark.dk)
Claus Manniche (claus.manniche@slb.regionsyddanmark.dk)
Birgit Juul-Kristensen (bjuul.kristensen@health.sdu.dk)

Version: 4 Date: 30 November 2011

Author's response to reviews:

I have made the requested major revisions: the abstract in the submission is identical with the manuscript. Regarding the minor revisions: I have moved the titles of the figures after the reference list, I have removed the vertical edges of the tables and I have diminished the white area in figure 1.

Kind regards
Inge Ris Hansen